Author's response to reviews

Title: Hexane fraction of Ardisia crispa Thunb A.DC root inhibits inflammation-induced angiogenesis

Authors:

Dayang Erna Zulaikha Awang Hamsin (dayangerna@hotmail.com)
Roslida Abdul Hamid (roslida@medic.upm.edu.my)
Latifah Saiful Yazan (latifah@medic.upm.edu.my)
Che Norma Mat Taib (norma@medic.upm.edu.my)
Looi Ting Yeong (joyce_ylt86@hotmail.com)

Version: 5 Date: 28 November 2012

Author's response to reviews: see over
Dear Editor,

I herewith attached the edited manuscript upon the comment by the editor to undergo the copyediting. The manuscript has been checked by a colleague of mine, whom is native-English speaker. The text of abstract as well as the rest of the manuscript was edited (in blue font) to differentiate with the original manuscript.